Author’s response to reviews

Title: Frequency of tuberculous and non tuberculous mycobacteria in HIV infected patients from Bogota (Colombia)

Authors:

Martha I Murcia (mimurcia@usa.net)
Jorge E Gomez-Marin (jegomezmarin@hotmail.com)
Fernando S Alvarado (falvarado@cfl.rr.com)
Jose G Bustillo (ibustill@hotmail.com)
Ellen Mendivelson (hospitalsimonbolivar@yahoo.com)
Bertha Gomez (berthaiss@latinmail.com)
Clara I Leon (cleon@hemagogus.ins.gov.co)
William A Triana (watrianr@hotmail.com)
Erwing A Vargas (vsar79@latinmail.com)
Edgar Rodriguez (e_rodriguez_beltran@altavista.com)

Version: 4 Date: 7 Nov 2001

Dear Editor BMC Infectious Diseases:

We modified our paper "Frequency of tuberculous and non tuberculous mycobacteria in HIV infected patients in Bogota (Colombia)" as requested.

a) We eliminate one of the titles
b) Dr. Martha Murcia-Aranguren should be cited with two- name family as is usual in Latin America, this was modified in author details
c) We added a conclusion section at the end of the article
We hope these changes will make now the paper suitable for publication.

Yours sincerely,

Jorge E. Gomez-Marin MD, PhD